Author’s response to reviews

Title: Tumor-associated macrophage-derived IL-6 and IL-8 enhance invasive activity of LoVo cells induced by PRL-3 in a KCNN4 channel-dependent manner

Authors:

Heyang Xu (xuheyang2012@gmail.com)
Wei Lai (laiweiv2001@126.com)
Yang Zhang (zyykq1202@foxmail.com)
Lu Liu (liulusums@163.com)
Xingxi Luo (Luo-sums@Tom.com)
Yujie Zeng (sumsyorky@21.cn)
Heng Wu (sumswuheng@yahoo.com.cn)
Qiusheng Lan (lansysu@foxmail.com)
Zhonghua Chu (sumschuzhonghua@hotmail.com)

Version: 6 Date: 9 April 2014

Author’s response to reviews:

In the revised manuscript, I have corrected the grammar mistakes with the help of professional language editing service of Edanz, and in the revised manuscript, the informed consent has been added in the methods section.